Author's response to reviews

Title: In vitro antibacterial, antioxidant and cytotoxic activity of acetone leaf extracts of nine under-investigated Fabaceae tree species leads to potentially useful extracts in animal health and productivity

Authors:

Jean Paul Dzoyem (jpdzoyem@yahoo.fr)
Lyndy J McGaw (lyndy.mcgaw@up.ac.za)
Jacobus N Eloff (kobus.eloff@up.ac.za)

Version: 2 Date: 18 January 2014

Author's response to reviews: see over
18 January 014

The Executive Editor
BMC Complementary and Alternative Medicine

Dear Tom

Manuscript on biological activity of under investigated Fabaceae trees

The two reviewers made a number of proposals. We have addressed all of these proposals. I have also read through the manuscript again and have made a number of style and grammar corrections and have also clarified some issues that may cause difficulties for some readers that are not experts in this area.

I have attached two documents. One has track changes and highlights to make it easier for the reviewers to check the changes and a clean final copy.

I trust that the manuscript will now be acceptable for publication.

With kind regards

Kobus Eloff

Prof JN Eloff
Leader Phytomedicine Programme